Author’s response to reviews

Title: A comparison of zero-profile anchored spacer (ROI-C) and plate fixation in 2-level noncontiguous anterior cervical discectomy and fusion- a retrospective study

Authors:

Zongyu Zhang (414316457@qq.com)

Yawei Li (20105232050@suda.edu.cn)

Weimin Jiang (jwmpaper@126.com)

Version: 2 Date: 29 Oct 2017

Author’s response to reviews:

Dear editor,

We would like to express our sincere appreciation for your careful reading and valuable comments to improve this paper. We have made the corrections according to the comments. The corrected part was marked in red.